Author's response to reviews

Title: Difficulties associated with out-patient management of drug abusers by general practitioners. A cross sectional survey of general practitioners with and without methadone patients in the french-speaking part of Switzerland.

Authors:

Anne AP Pelet (Anne.Pelet@hospvd.ch)
Jacques JB Besson (Jacques.Besson@inst.hospvd.ch)
Alain AP Pecoud (Alain.Pecoud@hospvd.ch)
Bernard BF Favrat (Bernard.Favrat@chuv.ch)

Version: 6 Date: 17 December 2005

Author's response to reviews:

Anne Pelet
Centre Saint-Martin
Rue Saint-Martin 7
1003 Lausanne
Switzerland
Fax: 0041 21 316 16 26

Lausanne, the 17th of December 2005

Manuscript: No 5419605646972778

Dear Editor,

Here is our answer regarding the format of our manuscript

Title and abstract: we ensured that they are now identical in the manuscript and in the submission system. In fact, the new version was not recorded.

Author list: we changed the suffix according to our manuscript

Tables: we changed the visibility of the borders of table. They are not generated with tabbed text, but are divided into cells.

With best regards,

Anne Pelet, M.D.